EO-1 Weekly Status Week of Oct 27 – Nov 2, 2006

Day of Year 300 - 306
Mission Day 2165 - 2171
EO-1 General:

There were 103 Data Collection Events (DCEs) scheduled this week.  Effort is underway to have the EO-1 website server physically moved to the EO-1 MOC.  When this occurs, the website will be down for a short period of time.  
The instruments underwent decontamination cycles as follows:

· HyperSpectral Imager (HSI) performed a deicing from 10/28/2006 at 22:05z (06-301) to 10/29/2006 at 12:55z (06-302). 

· Advanced Land Imager (ALI) performed an outgassing from 10/28/2006 at 22:15z (06-301) to 10/29/2006 at 13:05z (06-302).

Spacecraft Subsystems:

All spacecraft subsystems functioned nominally with the exception of the Command & Data Handling (C&DH) anomaly (see below).
C&DH 
Effort continued on developing workarounds for the Command and Data Handling Medium Speed Serial Port (MSSP) anomaly.  The MSSP anomaly prevents normal playback of recorded engineering data during ground network (GN) contacts.  Three workarounds are being pursued.  The first is to add ground stations to increase the number of real time contacts.  The second is to record playback data to the WARP at high rate for downlinking via the X-band science data link instead of the S-band telemetry communications link.  The third is a fail safe measure to recover recorded telemetry, following an anomaly, by inserting it into the real time data stream.  Because of its limiting effect on real time data recovery, this technique would not normally be used unless desired recorded telemetry cannot be recovered from the WARP for whatever reason.  
A WARP record test was performed during the last weekend’s outgassing period.  Five minutes (2 blocks) of data was recorded during a real time contact.  This data was later successfully downlinked from the spacecraft during an X band support.  Subsequent analysis of the downlinked data showed that the test was fully successful.  Further testing of this workaround will take place during the next outgassing period in about two weeks.  For the next test, it is planned to record up to 25 min of data per file on the WARP.  
Work is underway to test the third workaround.

Electrical Power Subsystem (EPS)
A solar array characterization test was performed on 10/29/2006 at 08:40z (06-302).
Instruments:

The instruments operated nominally this week. 

Technology Activities:
The EO-1 extended mission phase has been transformed into an on-orbit testbed for advanced technology and hyperspectral research.  The status of various validation efforts is as follows:

Sensor Web & Virtual Observatory Demonstrations
Work is proceeding in the Open Geospatial Consortium (OGC) Webservices (OWS) testbed-4 area.  A meeting has been arranged with Dr. Liping Di of George Mason Univ. who is both an ESTO AIST PI and a participant in the OGC OWS-4.  We are going to coordinate the use of his copy of Business Processing Execution Language (BPEL) to automatically sequence a set of science processing services to create a science product.  In our case, we will use Sensor Observation Service (SOS) to retrieve Level 1G Hyperion data and extract three of the 220 spectral bands and convert them to TIFF.  Then, Web Processing Service (WPS) will classify features within the image to locate a fire or some other event of interest.  Finally, Web Coverage Service (WCS) will put the features of interest on a map and deliver this product to a user.  This will be done in conjunction with a Sensor Planning Service (SPS) set of activities which will cause EO-1 to image the salient location.
Autonomous Sciencecraft Experiment (ASE)

ASE controlled EO-1 this week.  
Ground and Space Network
EOS Data and Operations System (EDOS), GSFC, MD, USA
There are no major problems to report.

Wallops Ground Station (WGS), Wallops Island, VA, USA 

There are no major problems to report.


DataLynx Ground Station (PF1), Poker Flat, AK, USA

There are no major problems to report.
Alaska Ground Station (AGS), Poker Flat, AK, USA 

AGS is currently “RED” for EO-1 X-Band.  AGS is expected to resume normal operations on November 3, 2006.


Svalbard Ground Station (SGS), Longyearbyen, Norway
There are no major problems to report this week.

Tracking and Data Relay Satellite System (TDRSS), White Sands, NM, USA

There are no major problems to report.

USGS Multi-Satellite Ground Station (LGS), Sioux Falls, SD, USA

LGS is now being scheduled only for contingencies.


Tasmanian Earth Resources Satellite Station, a.k.a. Hobart (HGS), Hobart, Australia

There are no major problems to report.  

Cordoba Ground Station (CGS), Cordoba, Argentina


No passes were scheduled this week. 

Upcoming Events:

· Lunar calibration on 11/06/2006 (06-310)

· Orbit Lowering maneuver on 11/07/2006 (06-311)
Imagery Status:

Scenes and Engineering Cals planned for week of Oct 27 – Nov 2, 2006                     103
Total scenes and engineering calibrations planned for entire mission – approximately 30,872
Total Scenes:  ALI scenes in the Level 0 archive            28,372 (as of Nov 2, 2006)

                        Hyperion scenes in the Level 0 archive    28,123
Publications and Presentations Status (as of 07-27-06):

341 publications 

234 external presentations

  54 articles and press releases
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